**Author’s response to reviews**

**Title:** Evaluation of the Korean Version of the Self-assessment of Nursing Informatics Competencies Scale

**Authors:**

Kyoungsan Seo (turtlerabbit@naver.com)

Yul Ha Min (yulhamin@gmail.com)

Seung-Hye Choi (hera0511@gmail.com)

Haeyoung Lee (im0202@cau.ac.kr)

**Version:** 4 **Date:** 09 Dec 2019

**Author’s response to reviews:**

We greatly appreciate your comments. We revised the manuscript carefully according to the reviewer’s comment.

1. Please include the full name of each ethics committee that approved your study in the Ethics approval and consent to participate section. Additionally, please be sure to clarify whether ethics approval and/or permission was needed from each university to conduct your study in this section. If you did not need ethics approval from each institution please confirm that this complies with national guidelines or the need for approval was waived.

: We added the full name of each ethics committee that approved our study in the Ethics approval.

When receiving the Ethics approval, the names of the schools to be studied was included in the dossier. So, no further ethics approval and/or permission was required from each university.

2. Please remove the response from reviewers from the file inventory, as it is no longer needed at this stage.

: We removed it.
3. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

: According to your comment, we uploaded the final form for publication.